Author's response to reviews

Title: Incidence of human cystic echinococcosis in Slovenia

Authors:

   Jernej JL Logar (jernej.logar@mf.uni-lj.si)
   Barbara BS Soba (barbara.soba@mf.uni-lj.si)
   Tadeja TK Kotar (tadejakotar@hotmail.com)

Version: 3 Date: 17 March 2008

Author's response to reviews:

Authors's responses to reviewers, version 2

Response to Dr. Thomas TR Romig
The language was corrected and spelling checked

Response to Dr Franz Josef Conraths

Major Revision

Following your suggestion, Figure 1 shows the distribution of all the patients serologically tested for CE in eastern and western regions of Slovenia. Figure 1 in bracket in the text was moved from the Result section to the Discussion section in the last sentence of the second paragraph.

Discretionary Revision

In line 7 of the Results section, the findings of IHA titres of 32 patients with cysts in the liver and of 2 patients with cysts in the lung are included.

Minor Revisions

We agree with you that our manuscript refers to serological testing; therefore, the title of the manuscript was changed to »Serological evidence for human cystic echinococcosis in Slovenia«.